Author's response to reviews

Title: Quality Standards for child and adolescent mental health in primary care

Authors:

Kapil Sayal (Kapil.Sayal@nottingham.ac.uk)
Myanthi Amarasinghe (jithani.amarasinghe@kcl.ac.uk)
Sarah Robotham (srobotham@btinternet.com)
Caroline Coope (caroline.coope@ucl.ac.uk)
Mark Ashworth (markashworth@nhs.net)
Crispin Day (crispin.1.day@kcl.ac.uk)
Andre Tylee (andre.tylee@kcl.ac.uk)
Emily Simonoff (emily.simonoff@kcl.ac.uk)

Version: 2 Date: 9 February 2012

Author's response to reviews:

Dear Eden,

As requested, please find 2 amendments to the manuscript (Page 6 paragraph 1) that confirm: 1) the name of the ethics committee that approved the study and 2) written informed consent for participation in the study was obtained from participants.

Thank you for proceeding with peer review.

Yours sincerely

Kapil Sayal